Author's response to reviews

Title: Community stigma endorsement and voluntary counseling and testing behavior and attitudes among female heads of household in Zambezia Province, Mozambique

Authors:

Abraham Mukolo (abraham.mukolo@vanderbilt.edu)
Meridith Blevins (meridith.blevins@vanderbilt.edu)
Bart Victor (bart.victor@owen.vanderbilt.edu)
Heather N Paulin (heather.n.paulin@vanderbilt.edu)
Lara M. E. Vaz (lmevaz@gmail.com)
Moshin Sidat (mmsidat@gmail.com)
Alfredo Vergara (avedoce@gmail.com)

Version: 4 Date: 22 August 2013

Author's response to reviews: see over
August 22, 2013

Jane M. Dalumpines  
c/o Jimmar Dizon  
Journal Editorial Office  
BioMed Central

Dear Jane M. Dalumpines and Jimmar Dizon

MS: 2065612729512881. Community stigma endorsement and voluntary counseling and testing behavior and attitudes among female heads of households in Zambézia Province, Mozambique

Thank you for your letter of July 31st, 2013 in which you invited us to resubmit a copyedited version of the manuscript that we submitted on July 29th, 2013.

Bethanie Rammer of BioMedEdits has thoroughly edited the manuscript (abstract, main text, tables and figures) for English-language usage, clarity and consistency. We hope that the manuscript format now complies with the journal’s specifications.

We look forward to the acceptance of the manuscript.

Sincerely,
Abraham Mukolo